Author’s response to reviews

Title: Ectopic Insulinoma: Case Report

Authors:

Mengqing Sun (sunmq1128@126.com)
Yaping Luo (luoyaping@live.com)
Yan You (youyan@pumch.cn)
Xianlin Han (hanxianlin@pumch.cn)
Yupei Zhao (zhao8028@263.com)

Version: 1 Date: 17 Sep 2019

Author’s response to reviews:

To Ziad Abbassi (Reviewer 1): Abstract line 13: Hypoglycemia
This spelling mistake has been corrected. Thank you for your advice.

To Awale Mohamed Abdullahi, M.D (Reviewer 3):
1. Title you can modify your title by saying Ectopic Insulinoma: Case Report
The title has been modified. Thank you for your advice.
2. Show us the two cases separate photos and write which patient you talking about?
The photos of the two patients are shown in the report (Figure 1-8). The figure legends have described them to be of Case 1 or Case 2 (Figure legends section, page 19-20). Thank you for your advice.